Author's response to reviews

Title: Medical Students' Choices of Specialty in The Gambia: The Need for Career Counselling

Authors:

Mustapha Bittaye (bittaye@yahoo.com)
Akin Tunde Ademola Odukogbe (akin_tundeodukogbe@yahoo.com)
Ousman Nyan (onyan@utg.edu.gm)
Bintou Jallow (bintoujallow@hotmail.com)
Akinyinka O Omigbodun (omigbodun@yahoo.com)

Version: 4 Date: 24 July 2012

Author's response to reviews: see over
Dear Editor,

We thank you most sincerely for your consideration of this paper for publication in your esteemed journal. We are very grateful to the reviewers for a great job done. Kindly find our itemised responses to the issues raised.

Reviewer I. Ian Wilson

We are grateful for your comments

Reviewer II. Jennifer Reath

We thank you for your comments

Reviewer III. Jean-Karl Soler

1. The confidence intervals have been added.
2. A univariate logistic regression analysis has been done for the factors that medical students considered contributory or not contributory to their choice of specialty as displayed in Table 3. A new Table 4 has been created to report the summary of the findings of the logistic regression analysis while the former Table 4 is now labelled as Table 5.
3. We have modified the narrative of the ‘Results’ to include the new analysis and to respond to the comments about interpretation of the contents of tables. The ‘Discussion’ section has also been modified accordingly. The areas that have been completely re-written are in a different ink colour.
4. We thank you for your useful suggestions.

With kind regards,